Author's response to reviews

Title: Computed tomography imaging of subpleural lipoma: a report of two cases.

Authors:

Christoph A Karlo (christoph.karlo@usz.ch)
Paul Stolzmann (paul.stolzmann@usz.ch)
Thomas Frauenfelder (thomas.frauenfelder@usz.ch)
Olivio F Donati (olivio.donati@usz.ch)
Sebastian Leschka (sebastian.leschka@usz.ch)

Version: 3 Date: 1 October 2010

Author's response to reviews: see over
Major revisions (we require the author to make these changes)

Abstract: please ensure that the Abstract is present in manuscript and matches exactly what is in the submission system. Please structure the abstract as follows: 'Introduction, Case presentation, Conclusions'. Please also ensure the patient identification (age, gender, ethnicity) is stated consistently in the manuscript and the abstracts on the manuscript and submission system. **DONE**

Author list: ['Olivio F Donati'] the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). **DONE**

Authors’ contributions: please ensure this is formatted into a paragraph rather than a list. Please also state that all authors have read and approved the final manuscript. **DONE**

Figure legends: Please provide a separate figure legend for each figure provided in the manuscript. **DONE**

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, 'The Title of My Prestigious Publication: A Scientific Study.' should be changed to 'The title of my prestigious publication: a scientific study' **DONE**

Title: please amend your title to include ': two case reports' **DONE**

Corresponding author: Please indicate the corresponding author on the title page, perhaps by the use of a key such as: `* Corresponding author' **DONE**

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) or job titles from the manuscript title page **DONE**

Email addresses: Please use authors' initials to indicate affiliation to email addresses **DONE**

Highlighting: Please remove the highlighting from the manuscript **DONE**

Structure: Please ensure the Competing interests section precedes the Consent section. Please also ensure the case reports sections start with the title 'Case presentations'. **DONE**

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. **DONE**

Thank you very much.

Sincerely,

C.Karlo